Author’s response to reviews

Title: Effectiveness and Safety of Ferric Carboxymaltose Compared to Iron Sucrose in Women with Iron Deficiency Anemia: Phase IV Clinical Trials

Authors:

Amreen Naqash (anaqash.scholar@kashmiruniversity.net)
Rifat Ara (rifatar88@gmail.com)
Ghulam Bader (gnbader@kashmiruniversity.ac.in)

Version: 1 Date: 22 Aug 2017

Author’s response to reviews:

BMWH-D-17-00245

Effectiveness and Safety of Ferric Carboxymaltose Compared to Iron Sucrose in Women with Iron Deficiency Anemia: Phase IV Clinical Trials

Amreen Naqash, B. Pharm, M. Pharm (Pharmacology); Rifat Ara, MBBS, MD (Obstetrics and Gynaecology); Ghulam N Bader, Ph. D., M. Pharm (Pharmacology)

BMC Women’s Health

Dear Editor,

Thank you for your feedback. We have made all necessary corrections that were mentioned by you. Please go ahead with further processing of the manuscript.

Warm regards

Ghulam N Bader
University of Kashmir